Author’s response to reviews

Title: Syphilitic hepatitis: A case report and review of the literature

Authors:
Jiaofeng Huang (huangjiaofeng@fjmu.edu.cn)
Su Lin (sumer5129@fjmu.edu.cn)
Mingfang Wang (wangmingfang@fjmu.edu.cn)
Bo Wan (wblu@163.com)
Yueyong Zhu (zhuyueyong@fjmu.edu.cn)

Version: 1 Date: 13 Apr 2019

Author’s response to reviews:

Dear reviewer and editor:

We would like to thank you for thoroughly reviewing our manuscript entitled ‘Syphilitic hepatitis: case report and review of the literature’. We acknowledge the reviewers’ comments and constructive suggestions very much, which are valuable for improving the quality of our manuscript. We hope, with these modifications and improvements based on your suggestions and the reviewers’ comments, the quality of our manuscript would meet the publication standard of BMC Gastroenterology. Here are our point-by-point responses:

Technical Comments:

1. Thank you for your careful work. We have supply the Figure1. We have checked the Declarations section in my manuscript, which now have full information.

Sincerely,

Yueyong Zhu, M.D.
Associated Professor of Medicine
Liver Disease Research Center
The First Affiliated Hospital of Fujian Medical University